Author’s response to reviews

Title: Retention in care and adherence to HIV and AIDS treatment in Anambra State Nigeria

Authors:
Chukwuma David Umeokonkwo (chukwumau@gmail.com)
Chima Ariel Onoka (chima.onoka@unn.edu.ng)
Pearl Adaoha Agu (pearlagu@yahoo.co.uk)
Edmund Ndudi Ossai (ossai_2@yahoo.co.uk)
Muhammad Shakir Balogun (muhammadsbalogun@gmail.com)
Lawrence Ogbonnaya (uluogbonnaya@yahoo.com)

Version: 1 Date: 27 May 2019

Author’s response to reviews:

Department of Community Medicine
Federal Teaching Hospital
Abakaliki
Ebonyi State
Nigeria
27th May 2019

Dr Cecillia Devoto
The Editor in Chief
BMC Infectious Diseases

Dear Devoto,

Point by Point Response to Reviewers Comments

We are delighted to submit the revised version of the manuscript titled “Retention in care and adherence to HIV and AIDS treatment in Anambra State Nigeria” by Chukwuma David Umeokonkwo, MBBS, MPH, FWACP; Chima Ariel Onoka; Pearl Adaoha Agu; Edmund Ndudi Ossai; Muhammad
The authors appreciate the detailed work of the reviewers at improving the quality of the manuscript through your diligent review, comments and suggested changes. The authors have carefully looked through the comments and suggestions and have made changes, provided more details to aid better understanding of the work.

Below is the point by point response of each of the comments and our responses. We have also provided a tracked changes and clean copy versions of the manuscript for easy review. We do hope that the work is now acceptable for publication. We available to provide more clarifications as might be needed. Thank you again for your time and professional input.

Yours sincerely

Dr Chukwuma Umeokonkwo